Author's response to reviews

Title: Healthcare workers and health care-associated infections: knowledge, attitudes, and behavior in emergency departments in Italy

Authors:

Cristiana Parmeggiani (cristiana.parmeggiani@virgilio.it)
Rossella Abbate (rossella.abbate@libero.it)
Paolo Marinelli (paolo.marinelli@unina2.it)
Italo F Angelillo (italof.angelillo@unina2.it)

Version: 2 Date: 16 December 2009

Author's response to reviews:

Thank you very much for having reviewed the manuscript “Health care workers and hospital infections: knowledge, attitudes, and behavior in emergency departments in Italy” for consideration for publication in BMC Infectious Diseases.

As suggested by the Associate Editor, we have corrected the minor grammatical errors and we have replaced in all manuscript the term "handwashing" by "hand hygiene measures".

My colleagues and I are most grateful for the extremely positive comments. We hope that these changes fully address the concerns raised therein and that have now brought the paper publishable in BMC Infectious Diseases.

Yours sincerely,

Italo F. Angelillo, DDS, MPH
Professor and Chairman